Author's response to reviews

Title: Rationale and design of PROSPECT-CONKO 004; a Prospective, Randomized trial Of Simultaneous Pancreatic cancer treatment with Enoxaparin and ChemoTherapy

Authors:

Hanno Riess (hanno.riess@charite.de)
Uwe Pelzer (uwe.pelzer@charite.de)
Andreas Hilbig (andreas.hilbig@charite.de)
Jens Stieler (jens.stieler@charite.de)
Bernhard Opitz (b.opitz@krankenhaus-halle-saale.de)
Theo Scholten (scholten@akh-hagen.de)
Doerte Kauschat-Brüning (doerte.kauschat-bruening@sanofi-ventris.com)
Peter Bramlage (peter.bramlage@mailbox.tu-dresden.de)
Bernd Doerken (bernd.doerken@charite.de)
Helmut Oettle (helmut.oettle@charite.de)

Version: 2 Date: 26 November 2008

Author's response to reviews: see over
Prof. Dr. Hanno Riess  
Charité Campus Virchow-Clini  
Medical Clinic Hematology / Oncology  
Augustenburger Platz 1  
13353 Berlin, Germany  

To:  
Natalie Pafitis MSc  
Senior Assistant Editor  
BMC-series Journals  
BioMed Central  
Middlesex House  
34-42 Cleveland Street  
London W1T 4LB  

**Revision Riess et al.**: “Rationale and design of PROSPECT-CONKO 004; a Prospective, Randomized trial Of Simultaneous Pancreatic cancer treatment with Enoxaparin and ChemoTherapy)”  

To the editors,  

Thank you for accepting the above mentioned article in principle for publication in *BMC Cancer*. As requested we have added the name of the ethics committee that granted approval to the methods section.  

We are looking forward to your final acceptance.  

Yours sincerely,  

Prof. Dr. Hanno Riess, MD